**Author's response to reviews**

**Title:** Diagnostic scope in out-of-hours primary care services in eight European countries: an observational study

**Authors:**

Linda AMJ Huibers (l.huibers@iq.umcn.nl)
Grete Moth (GRETE.MOTH@ALM.AU.DK)
Gunnar T Bondevik (Gunnar.Bondevik@isf.uib.no)
Janko Kersnik (janko.kersnik@ozg-kranj.si)
Carola A Huber (Carola.Huber@usz.ch)
Morten B Christensen (MBC@ALM.AU.DK)
Rüdiger Leutgeb (RLeutgeb@T-Online.de)
Armando M Casado (armandcasado.bcn.ics@gencat.cat)
Roy Remmen (roy.remmen@ua.ac.be)
Michel Wensing (M.Wensing@iq.umcn.nl)

**Version:** 4 **Date:** 28 April 2011

**Author's response to reviews:**

Dear Editor,

Herewith, we resubmit our manuscript entitled ‘Diagnostic scope in out-of-hours primary care services in eight European countries: an observational study’. We carefully revised the format as advised, and added the additional files at the end of the manuscript. We aimed to do this according to the format; please inform us if more changes need to be made.

Yours sincerely, on behalf of all authors,

Linda Huibers